Author's response to reviews

Title: Suicide or undetermined intent? - A register-based study of signs of misclassification

Authors:

Charlotte Björkenstam (charlotte.bjorkenstam@ki.se)
Ingemar Thiblin (ingemar.thiblin@surgsci.uu.se)
Lars-Age Johansson (lars_age.johansson@socialstyrelsen.se)
Peter Nordstrom (peter.nordstrom@ki.se)
Anna Fugelstad (anna.fugelstad@ki.se)
Johan Hallqvist (Johan.hallqvist@pubcare.uu.se)
Rickard Ljung (rickard.ljung@ki.se)

Version: 4
Date: 27 January 2014

Author's response to reviews:

Dear Sir/Madam,

We are pleased to submit an updated version the enclosed manuscript ”Suicide or undetermined intent? – A register-based study of signs of misclassification” for possible publication in Population Health Metrics.

We have added a multivariate analysis (table.5) and consequently added text to the method, result, and discussion sections.

This manuscript has not been submitted or published elsewhere and all the authors have approved of the enclosed manuscript. We do hope you will consider this for publication in Population Health Metrics.

Yours sincerely,

Charlotte Björkenstam